Author’s response to reviews

Title: Comparison of neonatal red cell transfusion reporting in neonatal intensive care units with blood product issue data: a validation study

Authors:
Jillian Patterson (jillian.patterson@sydney.edu.au)
Jennifer Bowen (jenny.bowen@health.nsw.gov.au)
Sally Francis (sally.francis@health.nsw.gov.au)
Jane Ford (jane.ford@sydney.edu.au)

Version: 2 Date: 09 Nov 2017

Author’s response to reviews:

Dear Dr Maria Zalm,

Thank you for reviewing our manuscript “Comparison of neonatal red cell transfusion reporting in neonatal intensive care units with blood product issue data: a validation study” (BPED-D-15-00177R1) for publication.

We have revised the manuscript in line with your requests, as detailed below.

If I can be of further assistance, please don’t hesitate to contact me.

Kind regards,

Jillian

Editor Comments:

1. Please upload a clean version of the manuscript without tracked changes.
   
   This has been done.

2. Please include all authors' names, e-mails and affiliations on the title page. Also, please indicate the corresponding author on the title page.

These details have been included on the title page
3. In the Authors Contributions please include the contributions of each author by their initials. Currently only three of the authors are specifically mentioned.

The contributions of all authors have been included in the author contribution statement (Page 13).

4. In accordance with BioMed Central editorial policies and formatting guidelines, all manuscript submissions to BMC Pediatrics must contain a Declarations section which includes the mandatory sub-sections listed below.

This section has been included (Page 13 of revised document)